Author's response to reviews

Title: Using routine clinical and administrative data to produce a dataset of attendances at Accident and Emergency Departments following self-harm

Authors:

Catherine Polling Dr (catherine.polling@kcl.ac.uk)
Alex Tulloch Dr (alex.tulloch@kcl.ac.uk)
Shikha Banerjee Miss (shikha.banerjee@kcl.ac.uk)
Sean Cross Dr (sean.cross@slam.nhs.uk)
Rina Dutta Dr (rina.dutta@kcl.ac.uk)
Paul Dargan Dr (paul.dargan@gstt.nhs.uk)
David Wood Dr (david.wood@gstt.nhs.uk)
Matthew Hotopf Prof (matthew.hotopf@kcl.ac.uk)

Version: 3
Date: 10 September 2014

Author's response to reviews:

Line numbers have been added to the manuscript.